Author's response to reviews


Authors:

Shedra Amy Snipes (sas84@psu.edu)
Sherrill L Sellers (sellersl@muohio.edu)
Adebola Odunlami Tafawa (aodunlam@hsph.harvard.edu)
Lisa A Cooper (lisa.cooper@jhmi.edu)
Julie C Fields (jfield13@jhmi.edu)
Vence L Bonham (bonhamv@mail.nih.gov)

Version: 5 Date: 29 July 2011

Author's response to reviews:

Dear Editor: Thank you for your attention to detail, catching the differences in electronic versions of the abstract. We have made the necessary changes, and the two abstracts now match. Again, we sincerely thank you for your detailed and thoughtful comments regarding our manuscript.